\begin{table}%t1
\caption{\label{cltable}Basic galaxy cluster information.}
%\centering
\par
\begin{tabular}{lccccccccccc}
\hline \hline \noalign{\smallskip}
Name &      $z$ & Ref.   &  $R_{\rm vir} $ & Ref. & $\sigma$ & Ref. & $L_{\rm X} $ & Ref. &  Dynamical state	& Ref. & $N$ \\
 &	& & $(h^{-1}~{\rm Mpc})$ & & $\rm km~s^{-1}$ & &	$(10^{44}~{\rm erg~s}^{-1})$ &  & &	&\\
\hline
MS~0451.6-0305     &      0.540&(1)	   &	  1.17&(5)  	&	1371&(7)  &    10.19&(10)	& relatively relaxed	&(12)&  4 \\	    
MS~2137.3-2353     &      0.313&(2)	   &	  1.95&(6) 	&	960&(9)   &    7.97&(10)  	& relaxed		&(13)&  5 \\
MS~1008.1-1224     &      0.3061&(3)	   &	  1.18&(5) 	&	1042&(8)  &    2.29&(10)  	& non-relaxed 		&(14)&  5 \\
Cl~0412-65         &      0.507&(4)	   &	  0.62&(5) 	&	700&(4)   &    0.16&(11)  	& 			&    &	2 \\
\hline
\end{tabular}
\tablebib {(1): \citet{D96}; (2): \citet{SMGMSWFH91}; (3): \citet{YEMAC98}; (4): \citet{DSPBC99}; (5): \citet{GM01}; (6):~\citet{ASFE03};
(7): \citet{CYEAGMP96}; (8): \citet{BGCYE99}; (9): \citet{KMPM95}; (10): \citet{LGHLA99}; (11): \citet{SEDCOSB97}; (12): \citet{Don96}; (13): \citet{JCBB05}; (14): \citet{AMWP02}.}\vspace{-2mm}
\tablefoot {The last column (N) gives the number of galaxies in each cluster that are used in the kinematical analysis.}\vspace{2mm}
\end{table}